\begin{table}[ht]
\centering
\begin{tabular}{lrrrrrrrr}
\sisetup{detect-weight=true,detect-inline-weight=math}
    Software & Version & Citation \\\hline
    \textbf{Trimmers} \\
    Atropos & 1.1.5 & This manuscript \\
    AdapterRemoval & 2.2.0 & \cite{schubert_adapterremoval_2016} \\    
    SeqPurge & ngs-bits commit 9e8d99d & \cite{sturm_seqpurge:_2016} \\    
    Skewer & 0.2.2 & \cite{jiang_skewer:_2014} \\ \\
    \textbf{Aligners} \\
    BWA & 0.7.15 & \cite{li_aligning_2013} \\
    bwa-meth & commit 1c530af & \cite{pedersen_fast_2014} \\
    Samtools & 1.4 & \cite{li_sequence_2009} \\
    STAR & 2.5.3a & \cite{dobin_star:_2013} \\ \\
    \textbf{Misc} \\
    ART & 151 & \cite{huang_art:_2012,jiang_skewer:_2014} \\
    Bedops & 2.4.26 & \cite{neph_bedops:_2012} \\
    SRA toolkit & 2.8.2-1 & \url{https://github.com/ncbi/sra-tools}
\end{tabular}
\caption{Descriptions of software used in the benchmark workflow.}\label{containers}
\end{table}
